Author's response to reviews

Title: The effect of osteoarthritis on functional outcome following hemiarthroplasty for femoral neck fracture: a prospective observational study

Authors:

Christoph K Boese (christoph.boese@uk-koeln.de)
Benjamin Bücking (buecking@med.uni-marburg.de)
Christopher Bliemel (bliemel@med.uni-marburg.de)
Steffen Ruchholtz (heuserpe@med.uni-marburg.de)
Michael Frink (frink@med.uni-marburg.de)
Philipp Lechler (lechler@med.uni-marburg.de)

Version: 5 Date: 10 August 2015

Author's response to reviews: see over
Dear Ms. Lopes de Sousa,

thank you for your kind email dated 02 July 2015. Both editorial points have been included in the revised manuscript.

If there remain further question please do not hesitate to contact me.

Sincerely,

Dr. Philipp Lechler

Point to point response:
- For any articles which disclose individual details and images, written informed consent for publication must have been obtained from the patient and a statement to this effect must be included in the manuscript. Please confirm whether consent to publish was obtained and include a statement to this effect in the manuscript.

  A: The statement concerning the approval for the publication of anonymized clinical data has been included in the materials and methods section.

- Please include the sentence "All authors read and approved the final manuscript" at the end of the Authors' Contribution section

  A: The sentence "All authors read and approved the final manuscript" has been included in the Authors’ Contribution section.